Author's response to reviews

Title: Anticipatory nausea in cyclical vomiting

Authors:

Fiona E McDonald (fionamcr@liverpool.ac.uk)
David R Fleisher (fleisherd@health.missouri.edu)

Version: 3 Date: 16 March 2005

Author's response to reviews:

BioMed Central Pediatrics
MS: 1320141810467648 - Anticipatory nausea in cyclical vomiting

Authors' Response to reviewer's comments #2

Thank you for the further comments relating to this manuscript. The points have now been addressed as follows:

Major
1) At the end of the case report, figures for the number of episodes over the past decade are given, thus demonstrating the progressive improvement in the condition.

2) The phrase "(although pre-existing anxiety could exacerbate ANV)" has been added to p.9, para 2.

Minor
On p.8, para 2, 'stigma' has been replaced by 'stigmatisation', as suggested.